Author’s response to reviews


Authors:

Gabrielle van der Velde (gabrielle.vandervelde@utoronto.ca)
Pierre Côté (pcote@uhnresearch.ca)
Ahmed M. Bayoumi (ahmed.bayoumi@utoronto.ca)
J. David Cassidy (dcassidy@uhnresearch.ca)
Eleanor Boyle (eboyle@uhnresearch.ca)
Heather M. Shearer (hshearer@uhnresearch.ca)
Maja Stupar (mstupar@uhnresearch.ca)
Craig Jacobs (cjacobs@uhnresearch.ca)
Carlo Ammendolia (cammendo@uhnresearch.ca)
Simon Carette (simon.carette@uhn.on.ca)
Maurits van Tulder (maurits.van.tulder@falw.vu.nl)

Version: 6 Date: 5 July 2011

Author’s response to reviews: see over
Formatting Changes Required (Final Revisions)


We have made the major revisions (formatting changes) and minor revisions (removal of additional files) as requested. Please find the list of changes below.

Best wishes,

Gabrielle van der Velde

Toronto, July 5, 2011

List of Revisions/ formatting changes:

1. Author List: the authors’ names are written as they are in the submission system.
2. Additional files were removed.
3. Minor formatting changes were made in the final proofs.